Author’s response to reviews

Title: Adverse events and in-hospital mortality: An analysis of all deaths in a Norwegian Health Trust during 2011

Authors:

Hans Flaatten (hans.flaatten@kir.uib.no)
Guttorm Brattebø (guttorm.brattebo@helse-bergen.no)
Bjørn Alme (baalme@me.com)
Kjersti Berge (kjersti.berge@helse-bergen.no)
Jan Rosland (JanHenrik.Rosland@ihelse.net)
Asgaut Viste (Asgaut.Viste@uib.no)
Bjørn Bertelsen (bjorn.inge.bertelsen@helse-bergen.no)
Stig Harthug (stig.harthug@helse-bergen.no)
Sidsel Aardahl (ruth.sidsel.fife.aardal@helse-bergen.no)

Version: 2 Date: 23 Jun 2017

Author’s response to reviews:

To the editor BMC Health Services Research!

Thanks for accepting our manuscript! We are pleased to see that both the reviewer and the editor find the manuscript of interest for the Journal!

I have now made the necessary and required correction: -Material and methods is changed to Methods,

- Table 3 in now corrected from erroneously Table 2, but this was correct in the text
- Some of the affiliations and e-mails of co-authors are updated.
- Alle authors have now accepted the final version of the manuscript and this is stated in the manuscript.

- Availability of data: here we used the standard text to explain why this is not possible to have public availability.

On behalf of the author group

Hans Flaatten